Author’s response to reviews

Title: Evaluation of microstructural changes in spinal cord of patients with degenerative cervical myelopathy by diffusion kurtosis imaging and investigate the correlation with JOA score

Authors:

ZhuoHang Liu (13894707172@163.com)
BingYang Bian (1241928390@qq.com)
Gang Wang (wanggangletter@126.com)
Cheukying Tian (Cheuk.Tang@mssm.edu)
ZhenShan Lv (lzspine@jlu.edu.cn)
ZhiQing Shao (390622965@qq.com)
Dan Li (13604430730@126.com)

Version: 2 Date: 20 Apr 2020

Author’s response to reviews:

Editor:
-1. Question: We note that Cheukying Tian is included on the title page, however, they are not referenced in the Authors' Contributions section. As such, please ensure that all authors are referenced in this section.
Answer: We appreciate the editor’s responsible attitude. Cheukying Tian's contribution was mentioned in the previous version of the manuscript, abbreviated to CT. With the consent of the author, we changed the abbreviation of his name to CYT.
-2. Question: Please ensure that all figures and tables are referenced in the main text of your manuscript.
Answer: I make sure that all figures and tables are quoted in our manuscript.

-3. Question: Please proofread and ensure that when you upload your revised submission it is in the final form for publication. Please remove any tracked changes or highlighting and include only a single clean copy of the manuscript. Should you wish to respond to these revision requests, please include the information in the designated input box only.
Answer: We make sure that the uploaded changes are published in the final form.

Reviewer 2: Dear authors,
The paper is more complete, adjustments are made to the form and the discussion was enriched. Thanks for your job.
Question: Thanks.